Author's response to reviews

Title: Secondary left ventricular injury with haemopericardium caused by a rib fracture after blunt chest trauma.

Authors:

Pankaj Kaul (pankaj.kaul@leedsth.nhs.uk)
Somshekar Ganti (gantisom@hotmail.com)
Graeme Macaulay (graeme_macaulay@hotmail.com)

Version: 3 Date: 20 March 2006

Author's response to reviews:

The following changes have been made to the article as requested:

1. Abstract section has the heading 'Abstract' now instead of 'Introduction'.
2. No references are cited in the abstract now.
3. 'Case Report' has been changed to 'Case report'.
4. The underlines have been removed from the section headings
5. Square brackets have been used for citing references in the text.
6. Bibliography has been renamed as references.
7. The reference list has been formatted in the style requested.